Author's response to reviews

Title: Clinically Relevant Quality Measures for Risk Factor Control in Primary Care: a retrospective cohort study

Authors:

Stefan Weiler (stefan.weiler@gmx.ch)
Armin Gemperli (armin.gemperli@unilu.ch)
Tinh-Hai Collet (tinh-hai.collet@chuv.ch)
Douglas C Bauer (dbauer@psg.ucsf.edu)
Lukas Zimmerli (lukas.zimmerli@usz.ch)
Jacques Cornuz (jacques.cornuz@chuv.ch)
Edouard Battegay (edouard.battegay@usz.ch)
Jean-Michel Gaspoz (jean-michel.gaspoz@hcuge.ch)
Eve A Kerr (ekerr@umich.edu)
Drahomir Aujesky (drahomir.aujesky@insel.ch)
Nicolas Rodondi (nicolas.rodondi@insel.ch)

Version: 3 Date: 14 February 2014

Author's response to reviews:

Dear Mr. Leoncio,

as requested we included a statement on patients' consent in the Methods section:

"The Institutional Review Boards at all sites decided informed consent from participants could be waived, as it was a retrospective study with anonymized data."

Kind regards

Nicolas Rodondi